# Lessons from the diagnosis and treatment of severe immune checkpoint inhibitor-associated pneumonia: a case report

**Fengxia Weng<sup>1</sup>**, **Jianhua Wei<sup>1</sup>**, **Ming Sang<sup>1</sup>**, **Xin Gao<sup>1</sup>**, **Ping Zhang<sup>1</sup> & Qinghui Fu\***,<sup>1</sup><sup>(D)</sup> <sup>1</sup>The Department of SICU, The First Affiliated Hospital, Zhejiang University School of Medicine, Hangzhou, Zhejiang Province, 310003, China

\*Author for correspondence: 2513107@zju.edu.cn

Herein, we report a case of an elderly male patient who underwent extended radical resection of cardiac carcinoma after regular chemotherapy combined with sintilimab (PD-1 monoclonal antibody) immunotherapy complicated with severe pneumonitis postoperatively. We performed several treatments for aspiration pneumonitis; however, the patient's pulmonary infection and oxygenation were not efficiently improved. The multidisciplinary team considered it an immune checkpoint inhibitor-associated pneumonitis after diagnosis and treatment and then modified the treatment regimen. The pulmonary inflammation was effectively controlled with improved oxygenation; the patient was gradually weaned from the ventilator and finally discharged. The possibility of immune checkpoint inhibitor-associated pneumonitis should be fully considered particularly for patients with a history of immunosuppressive therapy with clinical symptoms of severe pneumonitis.

**Plain language summary:** Pneumonia is well known. Immune pneumonia may be a new problem. It occurs in 2–5% of patients with immune therapy. It is a bad reaction with low incidence. If this disease is not treated in time, it will cause a relatively terrible result. The fatality rate can reach 12.8–22.7%. The most severe cases can be life threatening. At present, the reason for immune pneumonia is not clear. Some experts believe that it is related to immune change. Dyspnea, cough, fever and chest pain are symptoms of this disease. Although the incidence of immune pneumonia is very low, it should be noted. If you are on immunotherapy, consult your doctor when you feel unwell.

First draft submitted: 7 February 2023; Accepted for publication: 23 May 2023; Published online: 20 June 2023

Keywords: immune checkpoint inhibitor-associated pneumonitis • immune checkpoint inhibitors • immune system

Immune checkpoint inhibitors (ICIs), as novel immunotherapy, provide significant clinical benefits to patients with malignant tumors and prolong their overall survival [1]. However, immune system overactivation predisposes immune checkpoint inhibitor-associated pneumonitis (CIP) [2,3], which is a lung injury caused by ICIs and has various clinical, imaging and pathological manifestations. CIP has an incidence of 3–5% [4–6], severe incidence of 1.4% [7] and mortality of 12.8–22.7% [8–11]. Patients with CIP do not have typical symptoms and usually present with emerging or worsening dyspnea, cough, chest pain, fever and fatigue [12–14]; patients with severe CIP present with life-threatening respiratory symptoms that require ventilator support [15]. However, as many as a third of patients with CIP have atypical symptoms at onset, and clinicians can easily neglect therapeutic measures for CIP due to the presence of other clinical manifestations similar to aspiration pneumonitis [12]. The patient's hospital stay will be prolonged and mortality will be increased if the disease is not diagnosed accurately and in a timely manner [8]. The distinctiveness of this case is that the patient took timely and effective treatment measures for unimproved pulmonary infection after showing poor oxygenation in the early stage following the diagnosis of severe pneumonitis. Furthermore, severe CIP was diagnosed after discussion by the multidisciplinary team; the dose of hormone therapy was increased, and finally, the patient was effectively treated.









Figure 1. CT images of this patient at different stages of treatment. (A) The CT image on 19 June showing bilateral lung inflammation. (B) The CT image on 23 June showing that the bilateral lung inflammation was slightly improved compared with that on 19 June. The oxygenation of the patient was slightly improved, but there was more sputum in the trachea. (C) The CT image on 29 June showing patchy high-density shadows in the bilateral lung, progressed inflammation and deteriorated oxygenation again, indicating that the patient's condition was not improving even after adjusting antibiotics.

CT: Computed tomography.

# **Case description**

A 73-year-old male patient was diagnosed with gastric cancer in February 2022 and regularly treated with chemotherapy combined with sintilimab (PD-1 monoclonal antibody) immunotherapy. After preoperative preparation, he underwent laparoscopic extended radical resection of cardiac carcinoma (total gastrectomy + lymph node dissection + Roux-en-Y anastomosis anastomosis) + intestinal adhesiolysis on 2 June 2022. A nasogastric tube was used postoperatively for nutritional support therapy. He had an intermittent delirium attack, which was improved after using antipsychotics. On 19 June, the patient showed a progressive decrease in oxygen saturation, with an oxygen partial pressure of 54.9 mmHg; he was given oxygen inhalation via an oxygen storage bag mask at 10 l/min and transferred to an intensive care unit (ICU). After establishing the artificial airway, the ventilator was mechanically ventilated. Bronchoscopy revealed milky white fluid in the main airway and the right main bronchus that was blocking the bronchi, considering mistaken inhalation; thus, severe pneumonitis was diagnosed (Figure 1A).

However, there was considerable sputum in the airway, and daily bronchoscopic sputum suction was performed. On 23 June, his lung computed tomography (CT) showed bilateral lung inflammation, which was slightly improved compared with that on 19 June (Figure 1B). However, on 24 June, the patient's oxygenation became worse. Acute respiratory distress syndrome was caused by aspiration pneumonitis considering his medical history; thus, prone position ventilation was started.

On 27 June, the patient's sputum pathogen metagenome sequencing test report suggested *Enterococcus faecium*, *Candida albicans* and human herpes virus 1 type, with 99% confidence in identification. Thus, piperacillin sodium and tazobactam sodium combined with micafungin were continued. On 29 June, the patient's oxygenation further declined. Auscultation revealed low breath sounds in the right lower lung, and CT showed bilateral lower lung consolidation, predominantly in the right lung (Figure 1C); thus, prone position ventilation was given for 16 h. The patient had persistent airway hyperreactivity, barrel chest and a smoking history of >20 years, which was considered to be acute exacerbation of chronic obstructive pulmonary disease; thus, methylprednisolone 40 mg every 12 h was added. After the patient was treated with prone ventilation combined with steroids, his oxygenation was slightly improved with an oxygenation index up to 290 mmHg, but poor transient oxygenation still existed.



Figure 2. A CT image of the patient's lungs after treatment. (A) The computed tomography image on 20 July 2022 showing that the patient's bilateral lung inflammation was significantly improved and (B) predischarge computed tomography image on 17 August 2022.

The patient still required ventilator-assisted ventilation after adding antibiotics, with a ventilator pressure of 22 cm  $H_2O$ , a positive end-expiratory pressure of 5–8 cm  $H_2O$  and an oxygen concentration of 35–50%; his oxygenation and respiratory function did not improve significantly. Moreover, several consecutive CT imaging scans showed that there were no obvious or only slight improvements in the patient. Therefore, on 6 July, a multidisciplinary team was organized to discuss the case and enhance the patient's therapeutic efficiency. The patient had a history of therapy; thus, immune checkpoint inhibitor-associated pneumonia could not be excluded, and T-cell subset assessment was supplemented. The test results were as follows: total lymphocyte count: 1861, T cells (CD3<sup>+</sup>): 82.21% and T helper cells (CD3<sup>+</sup>, CD4<sup>+</sup>): 53.95%; this was treated with methylprednisolone 80 mg every 12 h. Considering the patient's long time of tracheal intubation and difficulty in weaning from the ventilator in the future, a tracheotomy was performed on 11 July. On 21 July, his lung CT showed an improvement in the lung infection compared with that before (Figure 2A), and oxygenation was also further improved. On 22 July, the patient was weaned from the ventilator, and oxygen inhalation was performed by mask at the pneumonectomy site. On 29 July, the patient was transferred from the ICU to a specialized ward for continuous treatment. On 17 August, his lung CT showed multiple inflammations in the bilateral lung, which was improved compared with the previous CT scan (Figure 2B). The laboratory parameters and ventilator-assisted ventilation settings are shown in Table 1.

Table 1.	The clinical tests and ventilator settings of this patient.						
Date (year 2022)	White blood count (10 <sup>9</sup> /l)	Neutrophils (%)	C reactive protein (mg/l)	Procalcitonin (ng/ml)	Ventilator pressure (cm H <sub>2</sub> O)	Positive end-expiratory pressure (cm H <sub>2</sub> O)	Oxygenation index
19 June	19.77	85.3	118.45	2.98	22	4	220
24 June	31.46	84.5	221	0.74	30	10	130
29 June	-	-	-	-	28	7	110
1 July	17.27	91.3	123.05	0.33	22	5	170
21 July	12.34	70.3	21.38	0.11	12	3	320
21 August	8.95	72.4	4.14	0.06	_	_	_

From 19 June to 29 June 2022, the patient's condition gradually progressed, and it was observed that oxygenation index decreased and inflammation-related indicators increased. After changing the treatment regimen, the patient's condition gradually improved from 1 July to 21 August 2022, with an increase in oxygenation index and a decrease in inflammation-related indicators.

## Discussion

Our patient was treated with a nasogastric tube for nutritional therapy after total gastrectomy due to an intermittent delirium attack, which could be a risk factor for aspiration pneumonitis [16,17]. Aspiration pneumonitis refers to adverse pulmonary consequences caused by gastric or oropharyngeal fluids or xenobiotics (e.g., ingested food pellets or fluids, mineral oil, salt or freshwater) that may contain bacteria and/or have low pH while entering the lower airways; chest imaging studies of aspiration pneumonitis show obstructive stenosis or foreign bodies in gravity-dependent areas or segmental involvement of the lungs [16,18]. It is clinically frequently accompanied by common symptoms of pneumonitis, comprising cough, fever, purulent sputum and dyspnea [19,20]. In our case, on transfer to ICU, the CT images showed tracheal patchy shadows and bronchoscopy showed milky fluid obstructing the main airway and right main bronchus; the initial diagnosis considered by the ICU doctor was severe pneumonitis caused by aspiration, that is, aspiration pneumonitis. During treatment, the patient's oxygenation did not improve as expected, and even after adding multiple antibiotics, the infection symptoms did not subside. After discussion with the multidisciplinary team, severe CIP was considered and diagnosis and treatment were adjusted accordingly. The conventional thinking of ICU doctors at the early stage of treatment for this patient resulted in an unclear diagnosis; however, the issues were found in time during the process and the treatment regimen was adjusted.

CIP is commonly observed when patients are treated with PD-1 or PD-L1 inhibitors [21–25]. A meta-analysis of 5038 patients from 19 clinical studies involving PD-1 and PD-L1 inhibitors showed a higher incidence of CIP and severe CIP in patients taking PD-1 inhibitors [26]. The onset of CIP differs from hours to 24 months after the first administration of ICIs, with a median onset of 2–3 months [3]. Severe CIP occurs 6 months after immunotherapy [27]; however, considering the interval and persistence of the immune response, CIP can occur at any time during immunotherapy. Therefore, close monitoring and follow-ups are important. Clinicians should be aware that patients with a history of treatment with ICIs may develop CIP [28].

Dyspnea (53%) and cough (15%) are the most common symptoms of CIP [29], and the imaging findings of CIP are varied and may demonstrate scattered or diffuse ground-glass opacities, patchy consolidation, interlobular septal thickening, reticular opacities, traction bronchiectasis and fibrous streak opacities in both lungs [30]. The few variances from other lung diseases or inflammations and the lack of typical symptoms can easily be confused with other diseases, resulting in neglect or even missed diagnosis of CIP [31,32].

In our patient, the bronchoscopy results were obtained on admission to ICU, indicating the presence of aspiration pneumonitis; moreover, its clinical manifestations and imaging findings were consistent with the diagnosis of aspiration pneumonitis. Thus, consequent treatment improved the oxygenation of the patient to some extent. The oxygenation of the patient decreased at the middle stage of treatment; therefore, prone ventilation and hormone therapy were given after considering acute respiratory distress syndrome and acute exacerbation of chronic obstructive pulmonary disease, which improved the symptoms of CIP to some extent [33,34]. After forming the diagnosis of CIP, the treatment regimen was further adjusted. The patient's condition greatly improved, he was gradually weaned from the ventilator and finally discharged. Since 2018, the China National Medical Products Administration has successively permitted the applications of >10 s such as PD-1 in China. However, at present, due to the short-term applications and lack of experience in the treatment, those similar complications require further summary and analysis. In clinical practices, ICU doctors tend to diagnose severe pulmonary infections similar to aspiration pneumonitis and ignore the history of immunosuppressive therapy. Therefore, we wanted to highlight this particular concern and help doctors to better diagnose CIP.

# Conclusion

It is difficult to clinically diagnose CIP. During the diagnosis and treatment of patients with a history of treatment with ICIs, the differential diagnosis of patients should be considered more carefully. If necessary, a relevant assessment of CIP should be performed as early as possible to allow patients to receive the most successful treatment.

### Summary points

- Immune checkpoint inhibitors, as novel immunotherapy, provide significant clinical benefits to patients with malignant tumors and prolong their overall survival.
- However, immune system overactivation predisposes immune checkpoint inhibitor-associated pneumonitis (CIP), which is a lung injury caused by immune checkpoint inhibitors and has various clinical, imaging and pathological manifestations.
- This report describes an elderly male patient who underwent extended radical resection of cardiac carcinoma after regular chemotherapy combined with sintilimab (PD-1 monoclonal antibody) immunotherapy complicated with severe pneumonitis.
- The patient's hospital stay will be prolonged and mortality will be increased if the disease is not diagnosed accurately and in a timely manner.
- The few variances from other lung diseases or inflammations and the lack of typical symptoms can easily be confused with other diseases, resulting in neglect or even missed diagnosis of CIP.
- In clinical practices, intensive care unit doctors tend to diagnose severe pulmonary infections similar to aspiration pneumonitis and ignore the history of immunosuppressive therapy.
- Therefore, we wanted to highlight this particular concern and help doctors to better diagnose CIP.

#### Author contributions

F Weng conceived this study. J Wei designed the study. F Weng and M Sang acquired and analyzed the data. X Gao and P Zhang contributed analysis tools. F Weng wrote the paper. Q Fu was of immense help in the preparation of the manuscript. All authors read and approved the final manuscript.

#### Financial & competing interests disclosure

This study was supported by the Zhejiang Medical and Health Science and Technology Project (ID: 2022499885). The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

No writing assistance was utilized in the production of this manuscript.

#### Ethical conduct of research

The authors state that they have obtained verbal and written informed consent from the patient/patients for the inclusion of their medical and treatment history within this case report.

#### **Open access**

This work is licensed under the Attribution-NonCommercial-NoDerivatives 4.0 Unported License. To view a copy of this license, visit http://creativecommons.org/licenses/by-nc-nd/4.0/

#### References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- 1. Chuzi S, Tavora F, Cruz M *et al.* Clinical features, diagnostic challenges, and management strategies in checkpoint inhibitor-related pneumonitis. *Cancer Manag. Res.* 9, 207–213 (2017).
- 2. Zhang Q, Tang L, Zhou Y, He W, Li W. Immune checkpoint inhibitor-associated pneumonitis in non-small cell lung cancer: current understanding in characteristics, diagnosis, and management. *Front. Immunol.* 12, doi: 10.3389/fimmu.2021.663986 (2021) (Online).
- Friedman CF, Proverbs-Singh TA, Postow MA. Treatment of the immune-related adverse effects of immune checkpoint inhibitors: a review. *IAMA Oncol.* 2(10), 1346–1353 (2016).
- •• Tells us how to treat the adverse effects of immune checkpoint inhibitors. It is very important.
- 4. Khunger M, Rakshit S, Pasupuleti V *et al.* Incidence of pneumonitis with use of programmed death 1 and programmed death-ligand 1 inhibitors in non-small cell lung cancer: a systematic review and meta-analysis of trials. *Chest* 152(2), 271–281 (2017).

- Naqash AR, Yang LV, Sanderlin EJ, Atwell DC, Walker PR. Interleukin-6 as one of the potential mediators of immune-related adverse events in non-small cell lung cancer patients treated with immune checkpoint blockade: evidence from a case report. *Acta Oncol.* 57(5), 705–708 (2018).
- 6. De Velasco G, Je Y, Bosse D *et al.* Comprehensive meta-analysis of key immune-related adverse events from CTLA-4 and PD-1/PD-L1 inhibitors in cancer patients. *Cancer Immunol. Res.* 5(4), 312–318 (2017).
- Provides a comprehensive review of the adverse effects of PD1 and CTLA-4 on tumor therapy.
- 7. Yu X, Zhang X, Yao T, Zhang Y, Zhang Y. Fatal adverse events associated with immune checkpoint inhibitors in non-small cell lung cancer: a systematic review and meta-analysis. *Front. Med. (Lausanne)* 8, doi: 10.3389/fmed.2021.627089 (2021) (Online).
- Wang DY, Salem JE, Cohen JV et al. Fatal toxic effects associated with immune checkpoint inhibitors: a systematic review and meta-analysis. JAMA Oncol. 4(12), 1721–1728 (2018).
- Suresh K, Voong KR, Shankar B et al. Pneumonitis in non-small cell lung cancer patients receiving immune checkpoint immunotherapy: incidence and risk factors. J. Thorac. Oncol. 13(12), 1930–1939 (2018).
- Tone M, Izumo T, Awano N *et al.* High mortality and poor treatment efficacy of immune checkpoint inhibitors in patients with severe grade checkpoint inhibitor pneumonitis in non-small cell lung cancer. *Thorac. Cancer.* 10(10), 2006–2012 (2019).
- 11. Cho JY, Kim J, Lee JS *et al.* Characteristics, incidence, and risk factors of immune checkpoint inhibitor-related pneumonitis in patients with non-small cell lung cancer. *Lung Cancer* 125, 150–156 (2018).
- 12. Wang H, Guo X, Zhou J *et al.* Clinical diagnosis and treatment of immune checkpoint inhibitor-associated pneumonitis. *Thorac. Cancer.* 11(1), 191–197 (2020).
- 13. Rashdan S, Minna JD, Gerber DE. Diagnosis and management of pulmonary toxicity associated with cancer immunotherapy. *Lancet Respir. Med.* 6(6), 472–478 (2018).
- 14. Suresh K, Naidoo J, Lin CT, Danoff S. Immune checkpoint immunotherapy for non-small cell lung cancer: benefits and pulmonary toxicities. *Chest* 154(6), 1416–1423 (2018).
- 15. Pozzessere C, Bouchaab H, Jumeau R et al. Relationship between pneumonitis induced by immune checkpoint inhibitors and the underlying parenchymal status: a retrospective study. ERJ Open Res. 6(1), doi: 10.1183/23120541.00165-2019 (2020) (Online).
- 16. Niederman MS, Cilloniz C. Aspiration pneumonia. Rev. Esp. Quimioter. 35(Suppl. 1), 73-77 (2022).
- 17. Lo WL, Leu HB, Yang MC, Wang DH, Hsu ML. Dysphagia and risk of aspiration pneumonia: a nonrandomized, pair-matched cohort study. J. Dent. Sci. 14(3), 241–247 (2019).
- 18. Ebihara S, Miyagi M, Otsubo Y, Sekiya H, Ebihara T. Aspiration Pneumonia: a key concept in pneumonia treatment. *Intern. Med.* 60(9), 1329–1330 (2021).
- 19. Kosutova P, Mikolka P. Aspiration syndromes and associated lung injury: incidence, pathophysiology and management. *Physiol. Res.* 70(Suppl. 4), S567–S583 (2021).
- Lascarrou JB, Lissonde F, Le Thuaut A et al. Antibiotic therapy in comatose mechanically ventilated patients following aspiration: differentiating pneumonia from pneumonitis. Crit. Care Med. 45(8), 1268–1275 (2017).
- 21. Delaunay M, Cadranel J, Lusque A *et al.* Immune-checkpoint inhibitors associated with interstitial lung disease in cancer patients. *Eur. Respir. J.* 50(2), doi: 10.1183/13993003.00050-2017 (2017) (Online).
- 22. Hodi FS, O'Day SJ, McDermott DF *et al.* Improved survival with ipilimumab in patients with metastatic melanoma. *N. Engl. J. Med.* 363(8), 711–723 (2010).
- Wolchok JD, Neyns B, Linette G et al. Ipilimumab monotherapy in patients with pretreated advanced melanoma: a randomised, double-blind, multicentre, phase 2, dose-ranging study. *Lancet Oncol.* 11(2), 155–164 (2010).
- 24. Buchbinder EI, Desai A. CTLA-4 and PD-1 pathways: similarities, differences, and implications of their inhibition. *Am. J. Clin. Oncol.* 39(1), 98–106 (2016).
- 25. Sun X, Roudi R, Dai T *et al.* Immune-related adverse events associated with programmed cell death protein-1 and programmed cell death ligand 1 inhibitors for non-small cell lung cancer: a PRISMA systematic review and meta-analysis. *BMC Cancer* 19(1), 558 (2019).
- 26. Ma K, Lu Y, Jiang S, Tang J, Li X, Zhang Y. The relative risk and incidence of immune checkpoint inhibitors related pneumonitis in patients with advanced cancer: a meta-analysis. *Front. Pharmacol.* 9, 1430 (2018).
- Makes us realize the seriousness and harm of immune-associated pneumonia.
- 27. Voong KR, Hazell SZ, Fu W *et al.* Relationship between prior radiotherapy and checkpoint-inhibitor pneumonitis in patients with advanced non-small-cell lung cancer. *Clin. Lung Cancer* 20(4), e470–e479 (2019).
- 28. Schneider BJ, Naidoo J, Santomasso BD *et al.* Management of immune-related adverse events in patients treated with immune checkpoint inhibitor therapy: ASCO guideline update. *J. Clin. Oncol.* 39(36), 4073–4126 (2021).
- 29. Wang H, Guo X, Zhou J et al. [Clinical diagnosis and treatment recommendations for the pneumonitis associated with immune checkpoint inhibitor]. Zhongguo Fei Ai Za Zhi 22(10), 621–626 (2019).
- 30. Naidoo J, Cottrell TR, Lipson EJ et al. Chronic immune checkpoint inhibitor pneumonitis. J. Immunother. Cancer 8(1), e000840 (2020).

- Naidoo J, Wang X, Woo KM *et al.* Pneumonitis in patients treated with anti-programmed death-1/programmed death ligand 1 therapy. *J. Clin. Oncol.* 35(7), 709–717 (2017).
- 32. Kawakami N, Saito H, Takahashi S *et al.* Airway disorders associated with immune checkpoint inhibitor therapy: two case reports and a systematic review. *Semin. Oncol.* 49(6), 439–455 (2022).
- 33. Trinh S, Le A, Gowani S, La-Beck NM. Management of immune-related adverse events associated with immune checkpoint inhibitor therapy: a minireview of current clinical guidelines. *Asia Pac. J. Oncol. Nurs.* 6(2), 154–160 (2019).
- 34. Beattie J, Rizvi H, Fuentes P *et al.* Success and failure of additional immune modulators in steroid-refractory/resistant pneumonitis related to immune checkpoint blockade. *J. Immunother. Cancer* 9(2), e001884 (2021).